

Geophysical Research Letters

RESEARCH LETTER

10.1029/2021GL094192

Key Points:

- Slip distribution and interactions of three recent subduction earthquakes in Mexico revealed from radar observations
- Increased probability of a nearfuture major earthquake around areas between the 2018 and 2020 events in Oaxaca
- Lowered chance of shallow coastal or offshore normal earthquakes but increased chance of inland normal ruptures

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Z. Li, zhenhong.li@chd.edu.cn

Citation:

Yu, C., Li, Z., & Song, C. (2021). Geodetic constraints on recent subduction earthquakes and future seismic hazards in the southwestern coast of Mexico. *Geophysical Research Letters*, 48, e2021GL094192. https://doi. org/10.1029/2021GL094192

Received 4 MAY 2021 Accepted 8 JUN 2021

© 2021. Changan University. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Geodetic Constraints on Recent Subduction Earthquakes and Future Seismic Hazards in the Southwestern Coast of Mexico

Chen Yu¹, Zhenhong Li^{1,2,3}, and Chuang Song¹

¹School of Engineering, Newcastle University, Newcastle upon Tyne, UK, ²College of Geological Engineering and Geomatics, Chang'an University, Xi'an, China, ³Key Laboratory of Western China's Mineral Resources and Geological Engineering, Ministry of Education, Xi'an, China

Abstract Three major subduction earthquakes occurred on March 20, 2012 (Mw 7.4), February 16, 2018 (Mw 7.2), and June 23, 2020 (Mw 7.4) in the southwestern coast of Mexico, which caused fatalities, casualties, considerable damage, and raised safety concerns about future seismic hazards. We use satellite geodetic observations to invert for the slip distributions of the three events and then investigate their interactions. Coulomb Failure Stress (CFS) induced by their slip both on surrounding thrust and normal faults are calculated. The positive CFS changes, along with the spatial-temporal seismicity evolution, approximate earthquake recurrence rate and interseismic coupling, collectively indicate an increased possibility of a near-future rupture around the areas between the 2018 and 2020 events in Oaxaca. Furthermore, there is a lowered chance of shallow coastal or offshore normal earthquakes but an increased chance of inland normal ruptures.

Plain Language Summary Large earthquakes result in stress changes, which can induce or retard regional seismic activity or even trigger other earthquakes. Therefore, understanding historical and recent earthquakes and their associated stress changes are crucial in the evaluation of future seismic hazards. One of the most popular and intuitive ways to relate past and future events is to calculate the Coulomb stress changes during an earthquake and a failure criterion can be established by considering the geometry of nearby faults and the stress changes induced by past events. This requires a detailed knowledge of the amount and direction of slip by which the main fault has slipped during its past ruptures. In this study, we use geodetic surface displacement measurements to invert for the slip distribution and the interactions of three subsequent subduction earthquakes in the southwestern coast of Mexico. We then evaluate the future seismic hazard of the region by considering Coulomb failure stress changes, spatial-temporal regional seismic activities, historic earthquake records and approximate earthquake recurrence rate. The results show increased chances of a near-future reverse rupture around Santa Catarina Juquila, Oaxaca and inland normal ruptures, but a lowered chance of shallow coastal or offshore normal ruptures.

1. Introduction

The southwestern Pacific coast of central Mexico suffers geohazards associated with frequent seismic events primarily due to the subduction of the Cocos plate beneath the North America plate at a rate of ~60 mm/ year (Figure 1). Three large subduction earthquakes (Mw > 7.0) struck the region consecutively on March 20, 2012 (Mw 7.4), February 16, 2018 (Mw 7.2), and June 23, 2020 (Mw 7.4). The 2012 event nucleated near Copala, Guerrero at a depth of ~20 km (Graham et al., 2014). The strongest aftershock reached Mw 6.0 and struck on 2 April according to the National Seismological Service (NSS) of Mexico. The 2018 event nucleated near Pinotepa Nacional (PN) in Oaxaca at a depth of ~25 km (Li et al., 2020), slightly to the west of the location of the 2012 event. Apart from these large earthquakes, Global Positioning System (GPS) measurements also revealed that Slow Slip Events (SSEs) were repeatedly occurring in the region approximately every 1.1 years (Vergnolle et al., 2010) which could be the triggering mechanism of the 2012 event (e.g., Graham et al., 2014). The 2020 event struck further east of Oaxaca near San Pedro Pochutla (SPP) at a depth of ~20 km. These three events collectively caused 12 deaths, 49 injuries, and damaged thousands of buildings.





Figure 1. Tectonic setting and seismic activities of the study region. Earthquakes with focal mechanism solutions by Sawires et al. (2019) from 1787 to 2018 and USGS from 2018 to 2020 are denoted as small colored dots for Mw < 7.0 events and beach balls for Mw \geq 7.0 events. Solid red lines are slab depth contours at every 20 km from the Slab 2.0 model (Hayes et al., 2018). PN: Pinotepa Nacional; PE: Puerto Escondido; SPP: San Pedro Pochutla; SCJ: Santa Catarina Juquila.

These subduction earthquakes play an important role in shaping the regional stress field and provide important insights into future seismic hazards. First, each of these events was preceded 17-55 years ago by a similar event that struck in the same area (Figure 1), namely the 1995 and 2012 events near Copala, the 1968 and 2018 events near PN and the 1965 and 2020 events east of SPP. It is therefore concerning that the most recent large subduction earthquake in the area between PN and SPP was only observed in 1978. Second, historical records reveal a notable seismic activity gap in the western part of the area affected by the 2012 event between Acapulco and Copala where no significant (Mw > 7.0) subduction events have occurred for at least 110 years (Kostoglodov et al., 2003). Kostoglodov et al. (1996) argued that this area had the potential to undergo an earthquake greater than Mw 8.0, but GPS measurements revealed that the SSEs might have released a significant part of the accumulated strain (Radiguet et al., 2012). Another seismic gap arises between the 2018 and 1978 events around Santa Catarina Juquila (SCJ) where the density of subduction events is low compared to its surroundings according to a detailed review of historical seismic activities by Sawires et al. (2019). Finally, apart from thrust-faulting earthquakes, the subduction of the Cocos plate has produced significant intraplate normal-faulting earthquakes occurring down-dip at some distance from the coupled subduction interface (e.g., the 1980 and 1973 events) as well as near the coast (e.g., the 1999 event east of SCJ). Apart from potential seismic hazards, the high angle normal faulting events such as the 1999 event may generate substantial tsunami and/or submarine landslide hazards. Whether, and to what extent, the recent large subduction events increased the possibility of these future hazards requires detailed investigation of their fault slip and the induced stress changes.

In this study, we utilize Radarsat-2 and Sentinel-1 Interferometric Synthetic-Aperture Radar (InSAR) observations (Figure 2) to invert for the detailed fault slip distributions of the three subduction earthquakes mentioned above. Combining the regional seismic activity from 2000 to 2020 and the Coulomb stress changes induced by the fault slip of the three events, we investigate the interaction of the closely located 2012 and 2018 events, the spatial-temporal evolution of the regional stress changes and the potential future seismic hazards.



Geophysical Research Letters



Figure 2. Observed, modeled and residual displacement maps of the three events. (a) Is for the 2012 event using Radarsat-2 images. (b and c) are respectively for the 2018 and 2020 events using Sentinel-1 images. (a1-a3): 20111213–20120411. (b1-b3): 20180205–20180217. (b4-b6): 20180215–20180221. (b7-b9): 20180214–20180226. (c1-c3): 20200619–20200625. (c4-c6): 20200622–20200628.

2. Data and Modeling Method

Line of Sight (LOS) surface displacements of Radarsat-2 and Sentinel-1 (Table S1) were generated using the GAMMA software and European Space Agency's precise orbits with the topographic phase contributions removed using the 3-arcsec (~90 m) Shuttle Radar Topography Mission digital elevation model (Farr et al., 2007). To reduce the long wavelength effects, we applied the tropospheric delay corrections provided by the Generic Atmospheric Correction Online Service for InSAR (Yu et al., 2018) and then subtracted a best-fit quadratic surface computed for each InSAR pair (Feng et al., 2015; Sreejith et al., 2016). A spatial multi-look filter and a Goldstein spectral filter (Goldstein & Werner, 1998) were applied to reduce phase noise, leading to an interferogram resolution of about 90 m. A minimum-cost-flow unwrapping method was used to retrieve the LOS displacements (Chen & Zebker, 2000). To reduce the spatial correlation of the deformation map due to its high resolution, we down-sampled the interferograms using a quadtree-based algorithm (Simons et al., 2002), resulting in 1,530, 9,630 and 5,210 data points respectively for the 2012, 2018, and 2020 events.

The resultant displacements were input to PSOKINV (Feng et al., 2013), assuming an elastic half-space (Okada, 1986). We first determined the fault geometry associated with each of the three events nonlinearly using a particle swarm optimization method (Feng et al., 2013), with the initial geometries set according to the slab 2.0 model (Hayes et al., 2018). We then divided the fault planes into 2 km by 2 km patches to allow the slip to vary spatially on each fault segment. An iterative grid search method (Feng et al., 2020; Fukahata & Wright, 2008) was then employed to estimate the spatially variable slip distribution. To reduce the uncertainty associated with the dip angles derived from the previous nonlinear inversion, we tested different values ranging from 5° to 35° with a 1° step and determined the optimal dip angle and slip distribution simultaneously by minimizing data misfits (Figure S1). We also implemented a Monte-Carlo analysis to quantify the uncertainties in the slip distributions using 100 different data sets, derived from the original interferograms randomly perturbed in a normal distribution about their original values using an a-priori standard deviation of 10 mm following Wright et al. (1999) (Figure S2). The resultant slip distributions (see





Figure 3

the supporting information) were then used to calculate the Coulomb Failure Stress (CFS) in Coulomb 3 (Lin & Stein, 2004; Toda et al., 2005) with an effective coefficient of friction of 0.4 (e.g., Graham et al., 2014; Mikumo et al., 1999). We calculated the CFS respectively for reverse receiver faults at 15 km using an average fault geometry from the slab 2.0 model, normal receiver faults at 15 km for shallow coastal events (using the geometry of the 1999 event) and normal receiver faults at 40 km for inland events (using the average geometry of the 1980, 1973, and 2017 events according to Sawires et al. (2019) for the 1980 and 1973 events and Melgar et al. (2018) for the 2017 event). We then investigated how the CFS varies with depth by calculating seven cross sections parallel or perpendicular to the trench as shown in Figure 3c2. For reverse receiver faults, we used the average strike and dip angles of the slab 2.0 model along each profile. For normal receiver faults, we used the geometry of the 1999 event for Profiles Y1 and Y2, and the average geometry of the 1980, 1973, and 2017 events for Profile Y3 (see Table S2 in the supporting information).

3. Slip Distributions and Interactions

The resultant slip distributions are shown in Figure 3a, where Figure 3a1 shows the 2012 event and Figure 3a2 shows collectively the three events. Most slips occurred at a depth between 18 to 25 km and the maximum slip reached ~7 m. There are three major patches where aftershocks concentrated after the 2012 event as shown in Figure 3b3 (black rectangles). Patches 1 and 2 clustered respectively at the northwest and southeast edges (along the strike direction) of the 2012 event slippage area (shown in Figure 4c along Profile Y2) suggesting intense stress concentration compared to its down-dip direction (Yamashita & Knopoff, 1987). Numerous small fractures around these tips or edges could interact through the stress fields perturbated by the mainshock and coalesced with one another because of the rapid increase of the stress-intensity factors as the cracks increase in size (McGarr, 1981). The occurrence of these aftershock clusters may therefore be explained by the sharpness of the rupture termination (Das & Henry, 2003). For example, the 1992 Landers earthquake had a sharp stress concentration at its southern end of the rupture where aftershocks clustered as opposed to its northern end. Such a phenomenon can also be observed in several other major earthquakes, for example, the Mw 8.1 1998 Antarctic Plate earthquake (Henry et al., 2000), the Mw 7.3 1992 Landers earthquake (Wald & Heaton, 1994) and the Mw 8.0 1985 Mexico earthquake (Mendoza & Hartzell, 1989). Furthermore, the along strike slip termination sharpness may reflect lateral variations of interseismic coupling (which increase gradually from the southeast to the northwest [e.g., Franco et al., 2012]) and/or irregularities in structural and geological properties of the fracture zones (Kostoglodov & Ponce, 1994).

The slip of the 2012 event resulted in substantial CFS increases as shown in Figure 3b1. However, we observed an area (~5 km wide) between Patches 2 and 3 where no co-seismic slip was observed and the increased CFS led to no apparent increase of seismicity compared to its surroundings (Figures 3b3 and 3b4). Both of its left and right sides behaved more like velocity-weakening regions and released stress via unstable slips, especially Patch 3 (right side) where the densest aftershock following the 2012 earthquake was observed despite its greater distance from the epicenter (Figure 3b1). This implies the area may have acted as a barrier that terminated the 2012 rupture. This barrier has sharp stress concentrations as evidenced by the dense aftershocks at its edges and may indicate a substantial geometrical discontinuity or a change in the strength properties of the rocks. Aki (1979) noted that this kind of sharp stress concentrator could be a barrier between twin earthquakes which describes well the relationship between the 2012 and 2018 events. The stress concentrated at the barrier could be released through nonelastic deformation or elastically in a future earthquake.

The termination of the 2012 rupture at its southeast end (Patch 2) led to large dynamic stress perturbations in the surrounding medium (Bhat et al., 2007) and a substantial increase of CFS which propagated further east to initiate numerous aftershocks, notably Patch 3. These aftershocks released the concentrated stress but the

Figure 3. (a) Slip distributions of the three earthquakes. (b) Δ Coulomb Failure Stress (CFS) on reverse receiver faults at 15 km depth and (c) Δ CFS on normal receiver faults respectively at 15 and 40 km depths. (d) shows the averaged Δ CFS in the black rectangle denoted in (b5) respectively for reverse receiver faults at 15 (red) and 25 km (purple), and normal receiver faults at 15 (blue) and 40 km (black). Blue dots in (a) and yellow dots in (b and c) are earthquakes during periods denoted in each sub-figure's caption from the Servicio Sismológico Nacional (SSN, http://www.ssn.unam.mx/). Co-seismic slip distributions are outlined by white dotted lines in (b1, b2, and b5). Areas affected by SSEs according to Colella et al. (2017), Radiguet et al. (2016), and Maubant et al. (2020) are enclosed by gray dotted lines in (b5). Dotted lines in (c2) are profiles used in Figure 4.





Figure 4. ΔCoulomb Failure Stress cross sections of trench-perpendicular profiles (a) and trench-parallel profiles (b) shown in Figure 3c2. (a1–a4) and (b1) are calculated on reverse receiver faults and (b2–b3) are on normal receiver faults (c) show the cumulative slip (black histograms) and number of cumulative earthquakes (lines) of each cross-profiles along Y2. Yellow dots are aftershocks from the Servicio Sismológico Nacional.

resulting slip further altered the stress field of its neighboring fault asperities (Rudnicki & Kanamori, 1981) which may be responsible for initiating the 2018 event. This event ruptured the areas down-dip of the after-shock cluster (Patch 3 in Figure 3a2) and the change in CFS (Δ CFS) predicted well the aftershock distribution northeast of the epicenter (Figure 3b4). Considering most aftershocks occurred above the subduction interface (Figure 4a2), we also investigated the CFS on the optimally oriented fault planes by considering the regional stress field (Pacheco & Singh, 2010) (Table S3) instead of using a fixed fault geometry (King et al., 1994). The resultant CFS change (Figure S3) has a similar pattern to that in Figure 3b5 and shows a clear relationship between the aftershock distribution and positive CFS changes. The lack of seismicity north of the 2012 event and northwest of the 2018 event can be explained by frequent SSEs (Figure 3b5) such as in 2011–2012 (Colella et al., 2017), 2014 (Radiguet et al., 2016), and 2017–2018 (Maubant et al., 2020). The maximum slip of the 2011–2012 SSE around SCJ reached ~0.1 m, which is not enough to accommodate the long-term plate convergence rate (Colella et al., 2017; Correa-Mora et al., 2008). Instead, the SSEs here are generally confined to deeper source regions which will further load the seismogenic zone upward (Colella et al., 2017).

The 2020 event ruptured the east of the SCJ region in an area with no significant CFS increase as shown in Figures 3b1 and 3b2 (cumulatively <0.1 bar), suggesting the static Δ CFS produced by the 2012 and 2018 events was unlikely the triggering mechanism. The seismic activity around the 2020 epicenter only

experienced a slight increase after the 2012 earthquake (Figure 3b1) and no substantial increase after the 2018 earthquake (Figure 3b2). However, the dynamic stress changes may have played a role in remote triggering as the passage of transient seismic waves produced by the surrounding events could have initiated a secondary mechanism (such as viscous flow, fluid flow, and afterslip) that promotes the 2020 rupture (Freed, 2005; Kilb et al., 2000). A similar delayed dynamic triggering mechanism has previously been proposed to be responsible for the triggering of the 1999 Hector Mine earthquake by the 1992 Landers earthquake (Kilb, 2003) and the 2002 Tonga deep earthquake sequence by a magnitude 7.6 earthquake occurred at a depth of 598 km (Tibi et al., 2003). No sharp slip termination and clear aftershock clusters were observed to date around the 2020 rupture probably due to the short period of observation (5 months), the pre-existence of stress shadows caused by the 1965 event (Figure 1) and/or SSEs such as in 2007 (Graham et al., 2016).

4. Future Seismic Hazards

One of the intuitive ways to investigate future seismic hazards is to calculate the Δ CFS induced by recent large earthquakes. Positive Δ CFS can raise the applied tectonic loading of a fault and move the fault closer to failure (Kilb et al., 2002). Numerous studies have demonstrated the correlation between positive Δ CFS and increased seismic activity or negative Δ CFS with zones of post-earthquake quiescence (e.g., Árnadóttir et al., 2004; King et al., 1994; Lin & Stein, 2004; Pope & Mooney, 2020). The CFS changes can also affect magmatic systems by triggering volcanic unrest and eruption (Biggs et al., 2013; Jenkins et al., 2021) and increase crustal permeability by channeling fluids to the seafloor (Bonini, 2019).

As shown in Figures 3b5 and 4a, the subduction earthquakes tended to increase the CFS, calculated on the reverse receiver faults, in areas immediately surrounding the major slipping patches, extending both upward and downward along the subduction slab (Profiles X1, X2, and X4) as well as to the southeast and northwest along the trench (Profile Y2). This overlaps the seismically dangerous Guerrero seismic gap (Figure 1) whose accumulated strain is comparable to at least Mw 8.0 and has not broken since 1911 (Kanamori et al., 1993). Profile X3 in Figure 4a3 also shows that most of the subduction slab near SCJ was loaded, especially the portion between 0-30 km (~0.3 bar increase). These positive Δ CFS could be responsible for the increase of the seismicity far from the mainshock epicenters such as the offshore events and those around SCJ. Most events are shallower than 30 km as shown in Figure 4a3, suggesting the region exhibits velocity-weakening frictional behavior and is capable of holding large earthquakes (Scholz, 1998). Radiguet et al. (2016) proposed that a 0.3–0.5 bar CFS increase induced by the SSE around the same area has triggered the 2014 Mw 7.3 Papanoa earthquake. Experiencing three subsequent positive Δ CFS perturbations, notably by ~0.24 bar at a depth of 15 km and ~0.28 bar at 25 km after the 2018 event and cumulatively ~0.3 bar at 25 km (Figure 3d), the fault segments around SCJ have been continuously brought closer to failure.

Another factor related to seismic hazard is the earthquake recurrence rate. The last recorded large subduction event around SCJ was 42 years ago (Sawires et al., 2019), which is spatially closer to the 2020 event although the uncertainty of its epicenter is high due to the lack of observations. Nishenko (1991) estimated the recurrence time of large subduction earthquakes in this region to be 54 ± 8 years based on historical earthquakes in 1870 (Ms 7.9), 1928 (Ms 7.8), and 1978 (Ms 7.8) which ruptured the same fault segment according to intensity and instrumental data. This estimate agrees well with the interseismic coupling inferred from GPS measurements (Correa-Mora et al., 2008; Scholz & Campos, 2012) and a review on recent subduction earthquakes by Sawires et al. (2019). These coherent seismic and geodetic observations indicate a fully coupled subduction interface around SCJ, which could result in repeating earthquakes of very similar size (Scholz & Campos, 2012). The recurrence rate of large subduction earthquakes inferred from the most recent data as reviewed by Sawires et al. (2019) in our study region increases from northwest to southeast with SCJ in the middle (17 years around Copala, 50 years around PN and 55 years around SPP as in Figure 1), suggesting a 42-year cycle is a sensible approximation in SCJ. Considering there were fewer subduction events around SCJ (Oaxaca gap in Figure 1) compared to its surroundings (Sawires et al., 2019), the continued strain accumulation on the subduction slab portion at this stage will no doubt raise safety concerns.

Seismic refraction and gravity data revealed a strong positive seismic primary wave velocity gradient from northwest to southeast along the middle America trench (Scholz & Campos, 2012; Valdes et al., 1986), resulting in a laterally segmented subduction plate with staggered rupture cycles. This explains the existence of the abovementioned barrier between Patches 2 and 3 (Figure 3b3), the lateral sharp slip termination and the seismic gaps. Conversely, considering the short temporal and spatial intervals between the 2012 and 2018 events, their ruptured asperities may have synchronized into a longer cycle in which both asperities would be ruptured simultaneously, breaking the barrier in between and resulting in a larger magnitude due to the increased rupture size. For example, this has been observed in the Kurile subduction zone where several neighboring Mw 7.5–7.8 earthquakes repeating every 30–50 years finally synchronized resulting in the 2011 Mw 9.1 earthquake (Hashimoto et al., 2009; Scholz & Campos, 2012). However, the synchronization could be obstructed due to the narrow downdip width of the seismogenic zone in this region (<80 km), given that Herrendörfer et al. (2015) proposed that cycle synchronizations tend to occur in regions with wider widths (>100 km).

We also calculated Δ CFS on normal receiver faults for coastal and inland earthquakes at 15 and 40 km depths (Figure 3c) and their CFS cross sections along Profiles Y1, Y2, and Y3 (Figure 4b). Overall, the recent subduction events generated stress shadows over coastal and offshore at shallow depths (0–28 km, Profile Y1 in Figure 4b2) which reduce the possibility of shallow normal earthquakes. Otherwise, those events could be potential tsunami generators (Sugawara et al., 2019) and trigger strike-slip earthquakes in the volcanic arc (Martínez-Díaz et al., 2004). The CFS along Profile Y2 shows complicated features where there are localized positive Δ CFS immediately below the ruptured faults and at shallow depths (0–10 km, Figure 4b3). We also observe 0.2–0.5 bar of CFS increases inland (Profile Y3 in Figure 4b4) which could pose a potential risk of future normal earthquakes. For example, Segou and Parsons (2018) proposed that an average of 0.5 bar CFS increase induced by the afterslip of the 2012 subduction earthquake was enough to trigger the 2017 inland normal earthquake (Figure 1). However, it is worth noting that the 2017 rupture may have already released the increased CFS caused by the 2012 event and generated sufficient stress shadows to compensate the stress increase caused by the 2018 event.

5. Conclusions

We used geodetic observations to invert for the slip distributions of three subduction earthquakes in the southwestern coast of Mexico along the Middle America Trench. Numerous aftershocks clustered at both the northwest and southeast slip patch edges of the 2012 event, but there is a lack of co-seismic slip and aftershocks, despite positive Δ CFS, between the 2012 and 2018 events where there may be a sharp barrier terminating the 2012 rupture at its southeast end. This termination triggered dense aftershocks further east which favored the subsequent 2018 rupture. Δ CFS calculated from the slip distributions of the three earthquakes predicted well the aftershock distribution and provided important insight into the region's future seismic hazard. The positive Δ CFS on reverse receiver faults induced subsequently by the three subduction earthquakes, along with evidence by investigating the regional subduction event recurrence rate, the interseismic coupling, and the level of the seismic activity, collectively indicate an increased possibility of a major future rupture around SCJ between the 2018 and 2020 events. There are also slight CFS increases over the Guerrero seismic gap, which has observed no major rupture (Mw > 7.0) for at least 110 years. Furthermore, Δ CFS calculated on normal receiver faults indicate a lowered chance of shallow coastal or offshore normal earthquakes occurring, but an increased chance of inland intraplate normal ruptures such as the destructive 2017 Puebla earthquake close to Mexico City.

Data Availability Statement

The Sentinel-1 data were obtained from ESA (https://scihub.copernicus.eu/dhus/#/home). The Radarsat-2 data were obtained from ESA through the third-party missions L-OADS online dissemination system (https://earth.esa.int/eogateway/catalog/radarsat-2-esa-archive). The seismic activity data were obtained from the Servicio Sismológico Nacional of Mexico (http://www.ssn.unam.mx/) and the United States Geological Survey (https://earthquake.usgs.gov/earthquakes/map/).



Acknowledgments

This research was funded by the National Natural Science Foundation of China through a major program (41941019) and the National Key Research and Development Program of China (2019YFC1509201). The work was also supported by the UK Natural Environment Research Council (NERC) through the Centre for the Observation and Modeling of Earthquakes, Volcanoes and Tectonics (COMET, come30001) and the LICS project (NE/ K010794/1).

References

- Aki, K. (1979). Characterization of barriers on an earthquake fault. *Journal of Geophysical Research*, 84, 6140–6148. https://doi.org/10.1029/JB084iB11p06140
- Árnadóttir, T., Geirsson, H., & Einarsson, P. (2004). Coseismic stress changes and crustal deformation on the Reykjanes Peninsula due to triggered earthquakes on 17 June 2000. Journal of Geophysical Research, 109. https://doi.org/10.1029/2004JB003130
- Bhat, H. S., Olives, M., Dmowska, R., & Rice, J. R. (2007). Role of fault branches in earthquake rupture dynamics. *Journal of Geophysical Research*, 112. https://doi.org/10.1029/2007JB005027
- Biggs, J., Chivers, M., & Hutchinson, M. C. (2013). Surface deformation and stress interactions during the 2007-2010 sequence of earthquake, dyke intrusion and eruption in northern Tanzania. *Geophysical Journal International*, 195, 16–26. https://doi.org/10.1093/gji/ ggt226
- Bonini, M. (2019). Seismic loading of fault-controlled fluid seepage systems by great subduction earthquakes. *Scientific Reports*, 9. https://doi.org/10.1038/s41598-019-47686-4
- Chen, C. W., & Zebker, H. A. (2000). Network approaches to two-dimensional phase unwrapping: Intractability and two new algorithms. *Journal of the Optical Society of America*. A, 17, 401–414. https://doi.org/10.1364/josaa.17.000401
- Colella, H. V., Sit, S. M., Brudzinski, M. R., Graham, S. E., DeMets, C., Holtkamp, S. G., et al. (2017). Seismicity rate increases associated with slow slip episodes prior to the 2012 Mw 7.4 Ometepec earthquake. *Earth and Planetary Science Letters*, 464, 35–45. https://doi. org/10.1016/j.epsl.2016.12.032
- Correa-Mora, F., Demets, C., Cabral-Cano, E., Marquez-Azua, B., & Diaz-Molina, O. (2008). Interplate coupling and transient slip along the subduction interface beneath Oaxaca, Mexico. *Geophysical Journal International*, 175, 269–290. https://doi.org/10.1111/j.1365-246X.2008.03910.x
- Das, S., & Henry, C. (2003). Spatial relation between main earthquake slip and its aftershock distribution. *Reviews of Geophysics*, 41. https://doi.org/10.1029/2002RG000119
- Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., et al. (2007). The shuttle radar topography mission. Reviews of Geophysics, 45, RG2004. https://doi.org/10.1029/2005RG000183
- Feng, G., Li, Z., Shan, X., Xu, B., & Du, Y. (2015). Source parameters of the 2014 Mw 6.1 South Napa earthquake estimated from the Sentinel 1A, COSMO-SkyMed and GPS data. *Tectonophysics*, 655, 139–146. https://doi.org/10.1016/j.tecto.2015.05.018
- Feng, W., Li, Z., Elliott, J. R., Fukushima, Y., Hoey, T., Singleton, A., et al. (2013). The 2011MW6.8 Burma earthquake: Fault constraints provided by multiple SAR techniques. *Geophysical Journal International*, 195, 650–660. https://doi.org/10.1093/gji/ggt254
- Feng, W., Samsonov, S., Qiu, Q., Wang, Y., Zhang, P., Li, T., & Zheng, W. (2020). Orthogonal fault rupture and rapid postseismic deformation following 2019 Ridgecrest, California, Earthquake sequence revealed from geodetic observations. *Geophysical Research Letters*, 47. https://doi.org/10.1029/2019GL086888
- Franco, A., Lasserre, C., Lyon-Caen, H., Kostoglodov, V., Molina, E., Guzman-Speziale, M., et al. (2012). Fault kinematics in northern Central America and coupling along the subduction interface of the Cocos Plate, from GPS data in Chiapas (Mexico), Guatemala and El Salvador. *Geophysical Journal International*, 189, 1223–1236. https://doi.org/10.1111/j.1365-246X.2012.05390.x
- Freed, A. M. (2005). Earthquake triggering by static, dynamic, and postseismic stress transfer. Annual Review of Earth and Planetary Sciences, 33, 335–367. https://doi.org/10.1146/annurev.earth.33.092203.122505
- Fukahata, Y., & Wright, T. J. (2008). A non-linear geodetic data inversion using ABIC for slip distribution on a fault with an unknown dip angle. Geophysical Journal International, 173, 353–364. https://doi.org/10.1111/j.1365-246X.2007.03713.x
- Goldstein, R. M., & Werner, C. L. (1998). Radar interferogram filtering for geophysical applications. *Geophysical Research Letters*, 25, 4035–4038. https://doi.org/10.1029/1998gl900033
- Graham, S., DeMets, C., Cabral-Cano, E., Kostoglodov, V., Rousset, B., Walpersdorf, A., et al. (2016). Slow Slip History for the MEXICO Subduction Zone: 2005 through 2011. Pure and Applied Geophysics, 173, 3445–3465. https://doi.org/10.1007/s00024-015-1211-x
- Graham, S. E., DeMets, C., Cabral-Cano, E., Kostoglodov, V., Walpersdorf, A., Cotte, N., et al. (2014). GPS constraints on the 2011-2012 Oaxaca slow slip event that preceded the 2012 March 20 Ometepec earthquake, Southern Mexico. *Geophysical Journal International*, 197, 1593–1607. https://doi.org/10.1093/gii/ggu019
- Hashimoto, C., Noda, A., Sagiya, T., & Matsu'Ura, M. (2009). Interplate seismogenic zones along the Kuril-Japan trench inferred from GPS data inversion. *Nature Geoscience*, 2, 141–144. https://doi.org/10.1038/ngeo421
- Hayes, G. P., Moore, G. L., Portner, D. E., Hearne, M., Flamme, H., Furtney, M., & Smoczyk, G. M. (2018). Slab2, a comprehensive subduction zone geometry model. *Science*, 362, 58–61. https://doi.org/10.1126/science.aat4723
- Henry, C., Das, S., & Woodhouse, J. H. (2000). The great March 25, 1998, Antarctic Plate earthquake: Moment tensor and rupture history. Journal of Geophysical Research: Solid Earth, 105, 16097–16118. https://doi.org/10.1029/2000jb900077
- Herrendörfer, R., Dinther, Van, Y., Gerya, T., & Dalguer, L. A. (2015). Earthquake supercycle in subduction zones controlled by the width of the seismogenic zone. *Nature Geoscience*, *8*, 471–474. https://doi.org/10.1038/NGEO2427
- Jenkins, A. P., Biggs, J., Rust, A. C., & Jara, R. L. (2021). A systematic approach to mapping regimes of earthquake-induced static stress changes acting on magmatic pathways. *Journal of Geophysical Research: Solid Earth*, 126. https://doi.org/10.1029/2020JB020242
- Kanamori, H., Jennings, P. C., Singh, S. K., & Astiz, L. (1993). Estimation of strong ground motions in Mexico City expected for large earthquakes in the Guerrero seismic gap. Bulletin of the Seismological Society of America, 83, 811–829.
- Kilb, D. (2003). A strong correlation between induced peak dynamic Coulomb stress change from the 1992 M 7.3 Landers, California, earthquake and the hypocenter of the 1999 M 7.1 Hector Mine, California, earthquake. *Journal of Geophysical Research*, 108(ESE 3–1). https://doi.org/10.1029/2001jb000678
- Kilb, D., Gomberg, J., & Bodin, P. (2000). Triggering of earthquake aftershocks by dynamic stresses. *Nature*, 408, 570–574. https://doi. org/10.1038/35046046
- Kilb, D., Gomberg, J., & Bodin, P. (2002). Aftershock triggering by complete Coulomb stress changes. Journal of Geophysical Research, 107(ESE 2–1). https://doi.org/10.1029/2001jb000202
- King, G. C. P., Stein, R. S., & Lin, J. (1994). Static stress changes and the triggering of earthquakes. Bulletin of the Seismological Society of America, 84, 935–953. https://doi.org/10.1016/0148-9062(95)94484-2
- Kostoglodov, V., Bandy, W., Domínguez, J., & Mena, M. (1996). Gravity and seismicity over the Guerrero seismic gap, Mexico. Geophysical Research Letters, 23, 3385–3388. https://doi.org/10.1029/96GL03159
- Kostoglodov, V., & Ponce, L. (1994). Relationship between subduction and seismicity in the Mexican part of the Middle America Trench. *Journal of Geophysical Research*, 99, 729–742. https://doi.org/10.1029/93JB01556

- Kostoglodov, V., Singh, S. K., Santiago, J. A., Franco, S. I., Larson, K. M., Lowry, A. R., & Bilham, R. (2003). A large silent earthquake in the Guerrero seismic gap, Mexico. *Geophysical Research Letters*, 30. https://doi.org/10.1029/2003GL017219
- Li, Y., Shan, X., Zhu, C., Qiao, X., Zhao, L., & Qu, C. (2020). Geodetic model of the 2018 Mw 7.2 Pinotepa, Mexico, earthquake inferred from INSAR and GPS data. *Bulletin of the Seismological Society of America*, 110, 1115–1124. https://doi.org/10.1785/0120190101
- Lin, J., & Stein, R. S. (2004). Stress triggering in thrust and subduction earthquakes and stress interaction between the southern San Andreas and nearby thrust and strike-slip faults. *Journal of Geophysical Research*, 109. https://doi.org/10.1029/2003jb002607
- Martínez-Díaz, J. J., Álvarez-Gómez, J. A., Benito, B., & Hernández, D. (2004). Triggering of destructive earthquakes in El Salvador. Geology, 32, 65–68. https://doi.org/10.1130/G20089.1
- Maubant, L., Pathier, E., Daout, S., Radiguet, M., Doin, M. P., Kazachkina, E., et al. (2020). Independent component analysis and parametric approach for source separation in InSAR time series at regional scale: Application to the 2017–2018 Slow Slip Event in Guerrero (Mexico). Journal of Geophysical Research: Solid Earth, 125. https://doi.org/10.1029/2019JB018187
- McGarr, A. (1981). Analysis of peak ground motion in terms of a model of inhomogeneous faulting. *Journal of Geophysical Research*, *86*, 3901–3912. https://doi.org/10.1029/JB086iB05p03901
- Melgar, D., Pérez-Campos, X., Ramirez-Guzman, L., Spica, Z., Espíndola, V. H., Hammond, W. C., & Cabral-Cano, E. (2018). Bend faulting at the edge of a Flat Slab: The 2017 Mw7.1 Puebla-Morelos, Mexico Earthquake. *Geophysical Research Letters*, 45, 2633–2641. https:// doi.org/10.1002/2017GL076895
- Mendoza, C., & Hartzell, S. H. (1989). Slip distribution of the 19 September 1985 Michoacan, Mexico, earthquake: Near-source and teleseismic constraints. Bulletin of the Seismological Society of America, 79, 655–669.
- Mikumo, T., Singh, S. K., & Santoyo, M. A. (1999). A possible stress interaction between large thrust and normal faulting earthquakes in the Mexican subduction zone. *Bulletin of the Seismological Society of America*, *89*, 1418–1427.
- Nishenko, S. P. (1991). Circum-Pacific Seismic Potential: 1989-1999. Pure and Applied Geophysics, 135, 169–259. https://doi. org/10.1007/978-3-0348-5639-3_2
- Okada, Y. (1986). Surface deformation due to shear and tensile faults in a half-space. International Journal of Rock Mechanics and Mining Science & Geomechanics Abstracts, 23, 128. https://doi.org/10.1016/0148-9062(86)90674-1
- Pacheco, J. F., & Singh, S. K. (2010). Seismicity and state of stress in Guerrero segment of the Mexican subduction zone. Journal of Geophysical Research: Solid Earth, 115. https://doi.org/10.1029/2009JB006453
- Pope, N., & Mooney, W. D. (2020). Coulomb stress models for the 2019 Ridgecrest, California earthquake sequence. *Tectonophysics*, 791, 228555. https://doi.org/10.1016/j.tecto.2020.228555
- Radiguet, M., Cotton, F., Vergnolle, M., Campillo, M., Walpersdorf, A., Cotte, N., & Kostoglodov, V. (2012). Slow slip events and strain accumulation in the Guerrero gap, Mexico. *Journal of Geophysical Research*, 117. https://doi.org/10.1029/2011JB008801

Radiguet, M., Perfettini, H., Cotte, N., Gualandi, A., Valette, B., Kostoglodov, V., et al. (2016). Triggering of the 2014 Mw 7.3 Papanoa earthquake by a slow slip event in Guerrero, Mexico. *Nature Geoscience*, *9*, 829–833. https://doi.org/10.1038/ngeo2817

- Rudnicki, J. W., & Kanamori, H. (1981). Effects of fault interaction on moment, stress drop, and strain energy release. Journal of Geophysical Research, 86, 1785–1793. https://doi.org/10.1029/JB086iB03p01785
- Sawires, R., Santoyo, M. A., Peláez, J. A., & Corona Fernández, R. D. (2019). An updated and unified earthquake catalog from 1787 to 2018 for seismic hazard assessment studies in Mexico. *Scientific Data*, 6. https://doi.org/10.1038/s41597-019-0234-z
- Scholz, C. H. (1998). Earthquakes and friction laws. Nature, 391, 37-42. https://doi.org/10.1038/34097
- Scholz, C. H., & Campos, J. (2012). The seismic coupling of subduction zones revisited. *Journal of Geophysical Research*, 117. https://doi. org/10.1029/2011JB009003
- Segou, M., & Parsons, T. (2018). Testing earthquake links in Mexico from 1978 to the 2017 M = 8.1 Chiapas and M = 7.1 Puebla Shocks. Geophysical Research Letters, 45, 708–714. https://doi.org/10.1002/2017GL076237
- Simons, M., Fialko, Y., & Rivera, L. (2002). Coseismic deformation from the 1999 Mw 7.1 Hector Mine, California, earthquake as inferred from InSAR and GPS observations. Bulletin of the Seismological Society of America, 92, 1390–1402. https://doi.org/10.1785/0120000933
- Sreejith, K. M., Sunil, P. S., Agrawal, R., Saji, A. P., Ramesh, D. S., & Rajawat, A. S. (2016). Coseismic and early postseismic deformation due to the 25 April 2015, Mw 7.8 Gorkha, Nepal, earthquake from InSAR and GPS measurements. *Geophysical Research Letters*, 43, 3160–3168. https://doi.org/10.1002/2016GL067907
- Sugawara, D., Yu, N. T., & Yen, J. Y. (2019). Estimating a tsunami source by sediment transport modeling: A primary attempt on a Historical/1867 normal-faulting tsunami in Northern Taiwan. Journal of Geophysical Research: Earth Surface, 124, 1675–1700. https://doi. org/10.1029/2018JF004831
- Tibi, R., Wiens, D. A., & Inoue, H. (2003). Remote triggering of deep earthquakes in the 2002 Tonga sequences. *Nature*, 424, 921–925. https://doi.org/10.1038/nature01903
- Toda, S., Stein, R. S., Richards-Dinger, K., & Bozkurt, S. B. (2005). Forecasting the evolution of seismicity in southern California: Animations built on earthquake stress transfer. *Journal of Geophysical Research: Solid Earth*, 110, 1–17. https://doi.org/10.1029/2004JB003415
 Valdes, C. M., Mooney, W. D., Singh, S. K., Meyer, R. P., Lomnitz, C., Luetgert, J. H., et al. (1986). Crustal structure of Oaxaca, Mexico, from
- seismic refraction measurements. Bulletin of the Seismological Society of America, 76, 547–563. Vergnolle, M., Walpersdorf, A., Kostoglodov, V., Tregoning, P., Santiago, J. A., Cotte, N., & Franco, S. I. (2010). Slow slip events in Mexico
- revised from the processing of 11 year GPS observations. Journal of Geophysical Research, 115. https://doi.org/10.1029/2009JB006852 Wald, D. J., & Heaton, T. H. (1994). Spatial and temporal distribution of slip for the 1992 Landers, California, earthquake. Bulletin of the
 - Seismological Society of America, 84, 668–691.
- Wright, T. J., Parsons, B. E., Jackson, J. A., Haynes, M., Fielding, E. J., England, P. C., & Clarke, P. J. (1999). Source parameters of the 1 October 1995 Dinar (Turkey) earthquake from SAR interferometry and seismic bodywave modeling. *Earth and Planetary Science Letters*, 172, 23–37. https://doi.org/10.1016/S0012-821X(99)00186-7
- Yamashita, T., & Knopoff, L. (1987). Models of aftershock occurrence. *Geophysical Journal of the Royal Astronomical Society*, 91, 13–26. https://doi.org/10.1111/j.1365-246X.1987.tb05210.x
- Yu, C., Li, Z., Penna, N. T., & Crippa, P. (2018). Generic atmospheric correction model for interferometric synthetic aperture radar observations. Journal of Geophysical Research: Solid Earth, 123, 9202–9222. https://doi.org/10.1029/2017JB015305